Author's response to reviews

Title: Survival of patients with metastatic breast cancer: twenty-year data from two SEER registries

Authors:

   Patricia Tai (ptai@scf.sk.ca)
   Edward Yu (Edward.Yu@lrcc.on.ca)
   Vincent Ving-Hung (conrvhgv@az.vub.ac.be)
   Gabor Cserni (cserni@freemail.hu)
   Georges Vlastos (Georges.Vlastos@hcuge.ch)

Version: 3 Date: 26 August 2004

Author's response to reviews:

To the Editor:

Some typos spotted in the manuscript:

1. Page 12
   The words "tumour" are changed to "tumor".

2. Page 18, Reference 12
   "Current Oncology" is changed to "Curr Oncol"

3. Page 19, Reference 23
   "Anal Quant Cytol Histol" is changed to "Anal Quant Cytol Histol"

4. Page 22, in Table 2, line 7,
   (5.8-10.8) should be changed to (3.5-7.9), and
   (3.8-14.4) should be changed to (4.0-4.8).

5. Page 24, in Table 4,
   "Brain turmors" is changed to "Brain tumors".

All the above typos are changed in the revised manuscript submitted.

Patricia Tai